Author’s response to reviews

Title: A three-year-old boy with X-linked adrenoleukodystrophy and congenital pulmonary adenomatoid malformation: a case report

Authors:

Ibrahim Abdulhamid (ihamid@med.wayne.edu)
Sermin Saadeh (SSaadeh@dmc.org)
Nedim Cakan (NCakan@med.wayne.edu)

Version: 2 Date: 6 December 2009

Author’s response to reviews:

I am responding to the last correspondence requesting minor changes.

These includes adding ethnic and country of origin of the child to the abstract and case description, removing figures and table from the manuscript and loading them separately.